\documentclass[letterpaper,12pt]{article}
\usepackage[utf8x]{inputenc}
\usepackage{amsmath}
\usepackage[letterpaper,margin=1in,includehead=true]{geometry}
\usepackage{comment}
\usepackage{graphicx}
\usepackage{fancyhdr}
\usepackage{enumerate}
\pagestyle{fancy}
\usepackage{color}
\usepackage{wrapfig}
%\usepackage{enumitem}
\usepackage{tikz, tikz-3dplot, pgfplots}
\usetikzlibrary{calc}
\pgfplotsset{compat=newest}



\title{\vspace*{-1.5in}\textbf{Math 2300-007: Quiz 5b}}
\date{}
\author{}


\begin{document}
\maketitle
%\underline{Math 1300-400: Quiz 1}
\lhead{Math 2300-007: Calculus II}
\chead{Feb., 16 2018}
\rhead{Quiz 5b}
\vspace{-.5in}
\noindent
{Name: \underline{\phantom{Aalsdkjflaskdjflaskjflksdjflkdsjf}}} \hfill{Score: \underline{\phantom{lksdjflkdsjf}}}
\vskip1cm
\noindent
\begin{enumerate}
\item Please rate the following aspects of my teaching:\\[12pt]
\begin{tabular}{l|ccccc}\hline
 Teaching Aspects&Lowest &&&& Highest\\
\hline
My explanations of key concepts &\;\;1\;\;&\;\;2\;\;&\;\;3\;\;&\;\;4\;\;&\;\;5\;\;\\
The examples I provide &\;\;1\;\;&\;\;2\;\;&\;\;3\;\;&\;\;4\;\;&\;\;5\;\;\\
My responses to student questions &\;\;1\;\;&\;\;2\;\;&\;\;3\;\;&\;\;4\;\;&\;\;5\;\;\\
My effectiveness at visiting all groups&\;\;1\;\;&\;\;2\;\;&\;\;3\;\;&\;\;4\;\;&\;\;5\;\;\\
My availability for assistance (office hrs, email, etc.) &\;\;1\;\;&\;\;2\;\;&\;\;3\;\;&\;\;4\;\;&\;\;5\;\;\\\hline
\end{tabular}\\[12pt]
\item My office hours are Mondays 3-5 (Math 340), Tuesdays 3-4 (Math 175), and Wednesdays 3-4 (Math 340). Circle the choices that apply.
\begin{enumerate}[(i)]
\item I am available during one of those times: \rule[-4pt]{1.5in}{.75pt}.
\item I am not available at any of those times.
\item I would like an office hour at \rule[-4pt]{1.5in}{.75pt}.\\[6pt]
\end{enumerate}
\item One thing I like about this section of Math 2300 is...\vskip1in
\item One thing I would like to see improved is...\vskip1in
\item Comments/Questions/Suggestions:\vskip3.5cm
\end{enumerate}

\end{document}
